Author’s response to reviews

Title: The Delirium and Population Health Informatics Cohort study protocol: ascertaining the determinants and outcomes from delirium in a whole population.

Authors:

Daniel Davis (daniel.davis@ucl.ac.uk)
Sarah Richardson (sarah.richardson1@newcastle.ac.uk)
Joanne Hornby (j.hornby@ucl.ac.uk)
Helen Bowden (h.bowden@ucl.ac.uk)
Katrin Hoffmann (k.hoffmann@ucl.ac.uk)
Maryse Weston-Clarke (maryse.weston-clarke@ucl.ac.uk)
Fenella Green (f.green@ucl.ac.uk)
Nishi Chaturvedi (n.chaturvedi@ucl.ac.uk)
Alun Hughes (alun.hughes@ucl.ac.uk)
Diana Kuh (d.kuh@ucl.ac.uk)
Elizabeth Sampson (e.sampson@ucl.ac.uk)
Ruth Mizoguchi (ruth.mizoguchi@nhs.net)
Khai Lee Cheah (khailee.cheah@nhs.net)
Melanie Romain (melanieromain@nhs.net)
Abhi Sinha (abhi.sinha@nhs.net)
Rodric Jenkin (rodricjenkin@nhs.net)
Carol Brayne (cb105@medschl.cam.ac.uk)
Alasdair MacLullich (a.maclullich@ed.ac.uk)

Version: 2 Date: 29 Jan 2018

Author’s response to reviews:
Thank you for the most recent input.

The box has been changed to Table 3 and referenced as such in the text.

AH and ES contributions have been detailed in the relevant section.

The corrected "change of author" forms will be emailed to Rien.